Old Dominion University
ODU Digital Commons

OES Faculty Publications

Ocean & Earth Sciences

2023

Acidification of Northeastern USA Lakes From Rising Anthropogenic-Sourced Atmospheric Carbon Dioxide and Its Effects on Aluminum Speciation

Karen H. Johannesson University of Massachusetts Boston

Jaxon Dii Horne University of Massachusetts Boston

Anant Misra University of Massachusetts Boston

Catherine Aliperta University of Massachusetts Boston

Orpheus V. Meletis University of Massachusetts Boston

See next page for additional authors Follow this and additional works at: https://digitalcommons.odu.edu/oeas_fac_pubs Part of the Environmental Sciences Commons, and the Fresh Water Studies Commons

Original Publication Citation

Johannesson, K. H., Horne, J. D., Misra, A., Aliperta, C., Meletis, O. V., Santore, R. C., White, C. D., Mavrommati, G., & Burdige, D. J. (2023). Acidification of northeastern USA lakes from rising anthropogenic-sourced atmospheric carbon dioxide and its effects on aluminum speciation. *Geophysical Research Letters*, *50*(22), 1-11, Article e2023GL104957. https://doi.org/10.1029/2023GL104957

This Article is brought to you for free and open access by the Ocean & Earth Sciences at ODU Digital Commons. It has been accepted for inclusion in OES Faculty Publications by an authorized administrator of ODU Digital Commons. For more information, please contact digitalcommons@odu.edu.

Authors

Karen H. Johannesson, Jaxon Dii Horne, Anant Misra, Catherine Aliperta, Orpheus V. Meletis, Robert C. Santore, Christopher D. White, Georgia Mavrommati, and David J. Burdige

This article is available at ODU Digital Commons: https://digitalcommons.odu.edu/oeas_fac_pubs/494

f



Geophysical Research Letters^{*}



10.1029/2023GL104957

Key Points:

- Rising atmospheric CO₂ from anthropogenic activities will lead to acidification of lakes in the northeastern USA during the 21st century
- Acidification will alter Al speciation and lower concentrations required to cause a sub-lethal response in 20% of the resident brook trout
- Anthropogenic-sourced CO₂ could reverse gains in lake water quality realized since implementation of the Clean Air Act

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

K. H. Johannesson, karen.johannesson@umb.edu

Citation:

Johannesson, K. H., Horne, J. D., Misra, A., Aliperta, C., Meletis, O. V., Santore, R. C., et al. (2023). Acidification of northeastern USA lakes from rising anthropogenic-sourced atmospheric carbon dioxide and its effects on aluminum speciation. *Geophysical Research Letters*, *50*, e2023GL104957. https://doi.org/10.1029/2023GL104957

Received 14 JUN 2023 Accepted 24 OCT 2023

Author Contributions:

Conceptualization: Karen H. Johannesson, Georgia Mavrommati, David J. Burdige Data curation: Karen H. Johannesson Formal analysis: Karen H. Johannesson, Jaxon Dii Horne, Anant Misra, Catherine Aliperta, Orpheus V. Meletis, Robert C. Santore, Christopher D. White, Georgia Mavrommati Funding acquisition: Karen H. Johannesson, David J. Burdige

© 2023 The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes.

Acidification Of Northeastern USA Lakes From Rising Anthropogenic-Sourced Atmospheric Carbon Dioxide and Its Effects on Aluminum Speciation

Karen H. Johannesson^{1,2}, Jaxon Dii Horne¹, Anant Misra¹, Catherine Aliperta¹, Orpheus V. Meletis¹, Robert C. Santore³, Christopher D. White¹, Georgia Mavrommati¹, and David J. Burdige⁴

¹School for the Environment, University of Massachusetts, Boston, MA, USA, ²Intercampus Marine Science Graduate Program, University of Massachusetts System, Boston, MA, USA, ³Winward Environmental, Syracuse, NY, USA, ⁴Department of Ocean and Earth Sciences, Old Dominion University, Norfolk, VA, USA

Abstract The impact of rising atmospheric CO_2 (p CO_2^{atm}) from anthropogenic activities on pH, dissolved inorganic carbon, carbonate mineral saturation, and aluminum (Al) speciation is evaluated for 18 northeastern USA lakes using polythermal, sliding activity reaction path models. p CO_2^{atm} was forced using two scenarios from the IPCC's Sixth Assessment Report in which p CO_2^{atm} attains either 600 or 1,100 ppm in 2,100. Results suggest pH will decrease 0.15 and 0.32 pH units, $a_{CO_3^2}$ - will decrease 24% and 49%, and $\Omega_{aragonite}$ will decrease 21% and 45%, respectively. These changes are of the same magnitude as those expected for the oceans. The effects of rising p CO_2^{atm} on sub-lethal 20% effect concentrations (i.e., EC20) of Al for brook trout (*Salvelinus fontinalis*) are evaluated with the biotic ligand model, which indicates Al toxicity effects will increase as pH decreases. These changes could reverse gains in water quality and fisheries health achieved since implementation of the Clean Air Act.

Plain Language Summary Fossil fuel consumption, deforestation, and changing land use are all contributing to rising CO_2 concentrations in the atmosphere. In addition to atmospheric warming and climate impacts, rising atmospheric CO_2 will also promote acidification of ocean surface waters, which is expected to have detrimental effects on marine ecosystems. Much less is known about how rising atmospheric CO_2 will impact terrestrial freshwaters such as lakes and streams. We developed reaction path models for 18 lakes from northeastern USA to investigate possible impacts of rising atmospheric CO_2 and temperature on these systems. The models indicate that acidification of these lakes could be of similar magnitude to ocean acidification. The saturation state of these lakes with respect to the mineral aragonite, which is important for calcifying organisms, will also decrease. Finally, acidification will alter aluminum (Al) speciation, which could augment toxic effects from this metal on important sport fish like brook trout, possibly reversing gains made in water quality since implementation of the Clean Air Act decreased the input of strong acids via acid rain.

1. Introduction

Acidification of the ocean from rising atmospheric CO_2 concentrations (p CO_2^{atm}) derived from anthropogenic activities, chief among which is the combustion of fossil fuels, is of concern for several reasons including ensuring the sustainability of important shellfish fisheries and marine ecosystems (Doney et al., 2009; Orr et al., 2005; Waldbusser et al., 2015). Despite efforts to understand and predict the impacts of human-driven rises in p CO_2^{atm} on ocean acidification (Balch et al., 2022; Millero et al., 2009), less effort has focused on the likelihood these processes will also lead to acidification of terrestrial, freshwater systems (e.g., Ninokawa & Ries, 2022; Phillips et al., 2015). Acidification of lakes and streams by increasing atmospheric CO_2 could be more marked and occur more rapidly than in marine systems as most terrestrial freshwaters exhibit lower alkalinities than the ocean and hence possess lower buffering capacities.

A complicating factor in the study of lake acidification by rising pCO_2^{atm} is that most lakes appear to be supersaturated with respect to CO_2 (e.g., Cole et al., 1994; Lazzarino et al., 2009; Raymond et al., 2013). Consequently, lakes are thought to be net sources of CO_2 to the atmosphere owing to the pCO_2 gradient between their surface waters and the overlying atmosphere (Alin & Johnson, 2007; Hanson et al., 2004). At the same time, direct measurements of lake surface water pCO_2 values (i.e., pCO_2^{water}) are relatively rare. Instead, most reported pCO_2^{water}



Johannesson

J. Burdige

Jaxon Dii Horne

J. Burdige

Investigation: Karen H. Johannesson

Methodology: Karen H. Johannesson,

Robert C. Santore, David J. Burdige

Project Administration: Karen H.

Resources: Karen H. Johannesson

Software: Karen H. Johannesson, Robert

C. Santore, Christopher D. White, David

Supervision: Karen H. Johannesson

Visualization: Karen H. Johannesson,

Writing – original draft: Karen H. Johannesson, David J. Burdige

Writing - review & editing: Karen H.

Johannesson, Jaxon Dii Horne, Anant

Misra, Robert C. Santore, Christopher D. White, Georgia Mavrommati, David

Validation: Karen H. Johannesson, Christopher D. White, David J. Burdige values are computed from coupled, in situ measurements of pH and alkalinity, pH and dissolved inorganic carbon (DIC) concentrations, or DIC and alkalinity using carbonate equilibria and geochemical software (e.g., Hunt et al., 2011). However, owing to the low accuracy and precision that commonly characterizes measurements of CO_2 -system parameters of many lake data sets, the computed pCO_2^{water} values are known to be highly uncertain and thus, difficult to evaluate as shown by direct comparisons of measured and calculated pCO_2^{water} values (Abril et al., 2015; Golub et al., 2017). Nonetheless, although lakes are commonly sources of CO_2 to the atmosphere, this feature will not inevitably shield them from the effects of rising atmospheric CO_2 -driven acidification (Phillips et al., 2015).

During the 1970s through the early 1990s meteoric deposition of strong acids such as sulfuric and nitric acid originating from the burning of high sulfur fossil fuels and automobile exhausts, respectively, led to the acidification of lakes and streams in the northeastern USA, Ontario, Quebec, and Atlantic Canada, as well as northern Europe (Driscoll, 1991; Likens et al., 1972; Schindler, 1988). The enactment of the Clean Air Act in the USA, the Eastern Canada Acid Rain Program in Canada, and similar legislation in Europe, helped to dramatically decrease SO₂ emissions from fossil fuel fired power plants, which has since led to the recovery of many previously acidified watersheds in these regions (Skjelkvåle et al., 2005; Stoddard et al., 1999; Strock et al., 2014). Although sulfuric acid deposition decreased substantially following the implementation of these environmental laws, nitric acid deposition appears to have remained relatively constant (Driscoll et al., 2003). These legislative efforts, however, did not explicitly address the increase of atmospheric CO₂ owing to fossil fuel consumption, deforestation, and land use change, or its potential impact on poorly buffered lakes and streams. In this contribution we conduct simple numerical experiments to investigate acidification of several oligotrophic lakes in northeastern USA that could occur between now and 2100 if pCO₂^{atm} attains a concentration of 600 or 1,100 ppm (e.g., Arias et al., 2021). Our modeling includes the impacts of rising atmospheric CO₂ and temperature on lake surface water pH, DIC species, and carbonate mineral saturation states. In addition, we examine the possible impact of CO₂-induced acidification on aluminum (AI) speciation in selected lakes including expected toxic effects of changing Al speciation on the important sport fish brook trout (Salvelinus fontinalis).

2. Methods

Major ion chemistry (Ca²⁺, Mg²⁺, Na⁺, K⁺, Cl⁻, SO₄²⁻), alkalinity, pH, dissolved silica, dissolved oxygen, water temperature, along with sampling dates, depths, and locations were obtained from the literature and online water quality databases maintained by individual US state agencies. The partial pressure of CO₂ in the atmosphere (pCO₂^{atm}) at the time of each lake's most recent sampling was estimated from the Mauna Loa Observatory record (NOAA, 2022) for the year sampled. For compiled information used in the modeling see Tables S1 and S2 in Supporting Information S1.

To investigate possible effects of rising pCO_2^{atm} on the studied lakes, we first developed equilibrium models for each lake's surface water using the compiled geochemical data. Subsequently, we simulated changes in surface water chemistry using polythermal, sliding activity/fugacity reaction path models to raise temperature (see O'Reilly et al., 2015) and pCO_2^{atm} from levels corresponding to the global estimate for the year when each lake was last sampled, up to a value of 1,100 ppm for 2100 as per scenario SSP5-8.5 of the IPCC's Sixth Assessment Report (Arias et al., 2021). Thus, in contrast to other investigations that present changes in surface water pH and carbonate chemistry relative to time (e.g., specific years, Phillips et al., 2015), our approach shows how the lakes could respond as a function of increasing temperature and pCO_2^{atm} up to a maximum of 1,100 ppm (scenario SSP5-8.5). If, however, rising pCO_2^{atm} more closely follows scenario SSP2-4.5 of the IPCC's Sixth Assessment Report where pCO_2^{atm} attains 600 ppm in 2100, then the model predicted changes in pH and carbonate chemistry can be estimated by inspection of the model plots assuming a final pCO_2^{atm} of 600 ppm (Figure 1). Although some studies indicate that pCO_2^{atm} may rise as high as 2,000 ppm by 2,300 (e.g., Caldeira & Wickett, 2003; Millero et al., 2009), we confine our assessments to changes during the 21st century. The modeling approach follows that of Phillips et al. (2015), which assumes that as pCO_2^{atm} increases, pCO_2^{water} will also increase proportionally to keep the lake-to-air CO₂ flux constant and maintain steady-state conditions in the lakes.

3. Results and Discussion

Equilibrium modeling indicates that most of the studied lakes are supersaturated with respect to atmospheric CO_2 (Figure S1 in Supporting Information S1). The only exceptions are Lake George, Nubanusit Lake, and Mendums





Figure 1. Predicted changes in pH owing to rising pCO_2^{atm} for 10 lakes from northeastern USA. Additional results for eight other lakes are presented in Figure S3 in Supporting Information S1.

Pond. Nevertheless, all the studied lakes are expected to become more acidic as pCO₂^{atm} increases, because pCO₂^{water} will also increase (e.g., McKinley et al., 2020; Phillips et al., 2015). Predicted pH changes in the studied lakes are shown in Figure 1 and Figure S3 in Supporting Information S1. Modeled changes in pH, the activity of carbonate species (i.e., $a_{H_2CO_3^*}$, $a_{HCO_3^-}$, and $a_{CO_3^2^-}$), and the saturation state of aragonite and calcite as a function of increasing pCO₂^{atm} for three lakes that span the range of pH buffering capacity of the studied lakes are presented in Figure 2. All results for the other 15 lakes investigated are shown in Figures S5–S10 in Supporting Information S1.

For scenario SSP5-8.5 of the IPCC's Sixth Assessment Report in which pCO_2^{atm} attains 1,100 ppm by 2100, the modeling indicates that pH would decrease between a low of 0.13 pH units in Mascoma Lake (New Hampshire) to as much as 1.31 pH units in Lake George (New York), with an average decrease across the 18 lakes of 0.32 pH units (Johannesson, 2023a). These pH decreases translate to increases in the activity of the hydrogen ion (i.e., a_{H^+}) of between 30% and 180%, with an average increase of 65% across all 18 lakes. If instead atmospheric CO₂ only attains a concentration of 600 ppm by 2100 (scenario SSP2-4.5), then the pH of surface waters of the studied lakes would decrease by between 0.05 and 0.83 pH units, with an average decrease of 0.15 pH units (Johannesson, 2023a). Hence, for scenario SSP2-4.2, the models indicate that, on average, a_{H^+} would increase by ca. 30% in the studied lakes. These changes are similar, albeit, slightly lower than predicted pH decreases for the ocean of 0.3–0.4 pH units and up to a 150% increase in a_{H^+} (Orr et al., 2005; Phillips et al., 2015).

There is a weak to moderate inverse correlation (i.e., $R^2 = 0.39$) between the buffering capacity of the 18 studied lakes and the model predicted changes in pH, which is statistically significant at the 95% confidence level (Figure S4 in Supporting Information S1). Hence, lower buffering capacity indicates a greater likelihood that rising atmospheric CO₂ will lead to a larger decrease in surface water pH. Nonetheless, testing this relationship requires careful monitoring of how pCO₂^{water}, pH, and alkalinity of lakes respond to changing pCO₂^{atm}.

The modeling demonstrates that carbonate anion activities, $a_{\rm CO_3^{2-}}$, would decrease by between 23.4% and 180%, and exhibit an average decrease of 49%, if pCO₂^{atm} rises according to scenario SSP5-8.5, whereas $a_{\rm CO_3^{2-}}$ would decrease between 8% and 146%, with an average decrease of 24%, assuming

that increases in pCO₂^{atm} more closely follow scenario SSP2-4.5. The average predicted decrease in $a_{CO_3^{2-}}$ for all 18 lakes is thus of the same magnitude as the expected 50% decrease in CO₃²⁻ concentrations in ocean surface waters (Orr et al., 2005; Phillips et al., 2015). If Nubanusit Lake and Mendums Pond (both in New Hampshire) are excluded, the average decrease in $a_{CO_3^{2-}}$ rises to 59% and 29% for scenarios SSP5-8.5 and SSP2-4.5, respectively. Decreases in the saturation state of the studied lakes with respect to aragonite and calcite are predicted to be nearly identical (Figure 2 and Figures S5–S10 in Supporting Information S1). Specifically, aragonite and calcite saturation states (i.e., log $Q/K = \log \Omega$) would decrease between approximately 20% to 179% for scenario SSP5-8.5 (average decrease ~45%), and 7%–145% for scenario SSP2-4.5 (average decrease ~21%).

Although the effect of rising atmospheric CO₂ on ocean acidification and hence marine calcifying organisms has elicited substantial concern (Andersson et al., 2008; Kawahata et al., 2019; Kleypas et al., 1999), the potential impact to freshwater calcifiers from increasing pCO₂^{atm} has only recently been addressed (Ninokawa & Ries, 2022). Rising atmospheric CO₂ increases the carbonic acid concentrations (H₂CO₃*) of surface waters according to Henry's Law (i.e., $a_{H_2CO_3} = K_{CO_2}pCO_2^{atm}$), which subsequently dissociates to HCO₃⁻ and H⁺, shifting the carbonate buffer system toward HCO₃⁻ as CO₃²⁻ reacts with H⁺ (Orr, 2011; Orr et al., 2005).

Geophysical Research Letters



Figure 2. Model results for three lakes that span the pH buffering capacity (i.e., β) of the 18 lakes studied (Table S1 in Supporting Information S1) that could result if pCO₂^{atm} increases to 1,100 ppm by 2100. Panels (a), (d), and (g) show the predicted changes in pH for Lake Champlain (Vermont—New York border), Moosehead Lake (central Maine), and Nubanusit Lake (southwestern New Hampshire). Panels (b), (e), and (h) present the log activity of H₂CO₃⁺, HCO₃⁻, and CO₃²⁻ that would accompany the decreasing pH, and panels (c), (f), and (i) demonstrates how carbonate mineral saturation states (i.e., log *Q/K* = log *Ω*) of each lake would change. The horizontal dashed lines in panels (c), (f), and (i) represent saturation for each mineral.

Because calcifying organisms like mollusks require the CO_3^{2-} anion to precipitate their shells, this decrease in the carbonate ion concentration forces these organisms to consume more energy to regulate their internal shell forming metabolic processes. The impact of CO_2 -induced acidification on carbonate mineral saturation states is particularly problematic for larval stages of calcifying organisms (Gledhill et al., 2015; Waldbusser et al., 2015). Unlike the ocean, freshwater systems are commonly undersaturated with respect to aragonite and calcite. Hence, freshwater calcifying organisms can tolerate substantially lower carbonate mineral saturation states than marine calcifiers (Ninokawa & Ries, 2022). Nevertheless, the much lower alkalinity and pH buffering capacity of freshwater systems compared to the ocean can still negatively impact freshwater calcifiers (Ninokawa & Ries, 2022), especially as rising atmospheric CO_2 drives these systems to progressively lower saturation states with respect to carbonate minerals (Beaune et al., 2018). Our modeling demonstrates that $a_{CO_3^2}$ - will decrease in the studied lakes as p CO_2^{atm} increases during the 21st century, pushing carbonate mineral saturation states to even more undersaturated conditions (Figure 2). The impact of these changes on indigenous calcifying organisms is currently unknown.

In addition to calcifiers, other freshwater organisms can also be adversely impacted by CO_2 -induced acidification and rising temperatures. For example, acidification can negatively impact freshwater crustaceans by lowering the activity of calcium ions in solution (Beaune et al., 2018; Tagliarolo, 2018). Moreover, increasing pCO₂^{water} has been shown to hamper growth rates and embryonic development of pink salmon (*Oncorhynchus gorbuscha*)





Figure 3. Aluminum speciation for three lakes from northeastern USA as a function of rising atmospheric CO_2 concentrations. In each panel Al-FA (bright blue curves) represents Al complexed with fulvic acid. Results for Lake Champlain are shown in panels (a), (b), and (c), Moosehead Lake in panels (d), (e) and (f), and Lake Sunapee (central New Hampshire) in panels (g), (h), and (i). Panels (a), (d), and (g) show predicted changes in Al species concentrations (as log [mol kg⁻¹]). Panels (b), (e), and (h) show the same results as percent total aluminum. Panels (c), (f), and (i) show percent Al versus pH for each lake. Dashed vertical black line in panels (c), (f), and (i) represents the current mean pH of surface waters in each lake, whereas the dashed blue vertical line to the left indicates pH in 2100 assuming scenario SSP5-8.5 of the IPCC's Sixth Assessment Report (Arias et al., 2021). Small red arrow shows the modeled acidification of these lakes.

during their freshwater development as well as egg hatchability, growth, and survival in brook trout, arctic char (*Salvelinus alpinus*), and Atlantic salmon (*Salmo salar*) among other economically important sport fish (Jagoe & Haines, 1983; Lacroix, 1985; Ou et al., 2015; Peterson et al., 1980).

Decreasing pH also commonly enhances concentrations of potentially toxic metals such as lead (Pb), mercury (Hg), and aluminum (Al), which together with pH stress, can have particularly negative impacts on freshwater biota (Cleveland et al., 1986; Parkhurst et al., 1990; Wang et al., 2016). Although amelioration of Al toxicity has been reported for lakes and streams in New York's Adirondack Mountains owing to substantial decreases in strong acid deposition (Baldigo et al., 2020; Michelena et al., 2016), future acidification of these freshwater environments by rising anthropogenic-sourced CO_2 could potentially reverse gains in these lakes and streams, as well as those of northern New England.

Aluminum speciation as a function of rising pCO_2^{atm} , and hence decreasing pH is shown in Figure 3 for Lake Champlain, Moosehead Lake, and Lake Sunapee, 3 of the 6 (out of 18) lakes we examined for which Al concentration data are available. The models indicate that in Moosehead Lake and Lake Sunapee Al chiefly occurs as complexes with naturally occurring dissolved organic matter (i.e., fulvic acid complexes), whereas the Al(OH)₄⁻ species predominates in Lake Champlain, followed closely by organic complexes (Figure 3). As pCO_2^{atm} rises, the fraction of Al complexed with fulvic acid should increase in all three lakes, although the





Figure 4. Predicted 20% effect concentrations (EC20) values of Al as a function of increasing pCO_2^{atm} , and thus decreasing surface water pH in panel (a) Lake Champlain, panel (b) Moosehead Lake, and panel (c) Lake Sunapee for brook trout (*Salvelinus fontinalis*) computed using the biotic ligand model (Santore et al., 2018). The predicted changes in pH in each lake are shown in Figure 1 as a function of rising pCO_2^{atm} .

relative increase is lowest in Lake Sunapee (Figure 3). Complexation of Al with dissolved organic ligands is thought to decrease the metal's toxicity (e.g., Driscoll et al., 1980; Fakhraei & Driscoll, 2015; Gensemer & Playle, 1999).

Nevertheless, the protective effect of organic complexation is compromised at circumneutral pH owing to the precipitation of amorphous to poorly crystalline, colloidal Al oxyhydroxides like amorphous gibbsite [i.e., $Al(OH)_{3(s)}$], which contributes to Al toxicity by binding to the surfaces of fish gills (Gensemer & Playle, 1999; Gensemer et al., 2018; Poléo et al., 1997). Lake Champlain and Moosehead Lake are saturated with respect to gibbsite and will remain so as pCO₂^{atm} increases during the 21st century, whereas Lake Sunapee is slightly undersaturated with respect to gibbsite (Figure S11 in Supporting Information S1).

Speciation modeling further demonstrates that even though Lake Sunapee has the lowest reported Al concentrations of the studied lakes, the concentration of positively charged inorganic Al species [i.e., Al^{3+} , $Al(OH)^{2+}$, and $Al(OH)_2^+$] occur at substantially higher concentrations than in either Lake Champlain or Moosehead Lake (Figure 3 and Figure S12 in Supporting Information S1). Specifically, the sum of the concentrations of these species is 36-fold higher in Lake Sunapee compared to Lake Champlain, and 8 times higher in Lake Sunapee than in Moosehead Lake. These positively charged ionic species can adsorb onto negatively charged brachial surfaces within fish gills leading to toxic osmoregulatory effects (e.g., Gensemer & Playle, 1999; Teien et al., 2006).

The biotic ligand model (BLM; Santore et al., 2018) predicts that the amount of Al complexed to the biotic ligand associated with brook trout in Lake Sunapee (15.8 nmol gw^{-1}) is 3-fold higher than in Lake Champlain (5.2 nmol gw^{-1}) and 3.5 times greater than in Moosehead Lake (4.4 nmol gw⁻¹), indicating a greater threat from Al in Lake Sunapee. Moreover, the BLM reveals that the 20% effect concentrations (EC20) for brook trout will decrease dramatically in these lakes as pCO₂^{atm} increases and the lakes acidify (Figure 4). EC20 values reflect the total Al concentration in each lake that will cause a sub-lethal response in 20% of the resident brook trout population (Wright & Welbourn, 2002). Hence, the Al concentration required to cause sub-lethal effects in 20% of the resident brook trout population will decrease in each lake as pCO₂^{atm} increases and pH decreases. Consequently, if total Al concentrations remain approximately constant in each lake, greater percentages of the resident brook trout populations would experience the effects of Al toxicity as the lakes acidify. The effects of Al toxicity to brook trout are currently greatest in Lake Sunapee and will remain so as pCO2 atm increases and the lake acidifies during the 21st century.

Most of the toxic effects from Al on brook trout in these lakes will be from precipitated Al oxyhydroxides like amorphous gibbsite (Figure S13 in Supporting Information S1). The relative percent of precipitated Al oxyhydroxides contributing to toxic effects (i.e., EC20) in brook trout is predicted

to decrease from 85.3%-81.9% of the total Al in Lake Champlain, 81.2%-76.8% of the total Al in Moosehead Lake, and 63.9%-56.3% of the total Al in Lake Sunapee as pCO₂^{atm} increases and these lakes acidify over the 21st century. Aluminum oxyhydroxides may form by direct precipitation from the saturated to nearly saturated lake waters, or within fish gills when acidic to circumneutral pH lake waters with relatively high Al concentrations encounter the more alkaline environment characteristic of fish gills (e.g., Gensemer & Playle, 1999; Gensemer et al., 2018; Playle & Wood, 1990). Indeed, the positive surface charge of Al oxyhydroxides (i.e., isoelectric point for gibbsite is ~9; Drever, 1988) will facilitate their adsorption onto the negatively charged gill sites, leading to toxic respiratory effects (Gensemer & Playle, 1999; Teien et al., 2006; Wilkinson et al., 1993).

19448007, 2023, 22, Downlo

from

https://agupubs.onlinelibrary.wiley

/10.1029/2023GL104957 by Old Dominion

University, Wiley Online Library on [09/01/2024].

. See

on Wiley Online Library for rules

;OA

ned by the applicable Creativ

Therefore, Al oxyhydroxide precipitates represent the greatest Al toxicity threat to brook trout in these lakes (Santore et al., 2018). Aluminum stress on brook trout may also be exacerbated as rising lake water temperatures decrease the solubility of dissolved oxygen, which is necessary for aerobic respiration (e.g., Jane et al., 2021).

Brook trout spawn in the autumn and require well aerated sandy and gravel-rich substrates to build their spawning beds. As such, they commonly migrate up streams from lakes to spawn. During the spring runoff, stream pH can attain values as low as 4 (Burns et al., 2019; Fakhraei & Driscoll, 2015). Many of the streams used by brook trout for spawning in northern New England and New York are also poorly buffered and can exhibit negative acid neutralizing capacities (e.g., Fuss et al., 2015; Gbondo-Tugbawa & Driscoll, 2003; Hall et al., 1985). Consequently, Al toxicity will be most dire for brook trout fry shortly after hatching in the spring when they are most vulnerable to the combined harmful effects of acidity and dissolved and precipitated Al species (e.g., Cleveland et al., 1986, 1989, 1991; Ingersoll et al., 1990; Mount et al., 1990; Wood et al., 1990). Additional research should thus focus on pCO₂^{atm} driven acidification of lakes and streams in the northeastern USA in combination with study of changing Al speciation and that of other metals/metalloid and their toxic effect on indigenous biota.

4. Conclusions

Rising pCO₂^{atm} resulting from anthropogenic activities will lead to acidification of the oceans and freshwater lakes during the 21st century. Reaction path models for lakes from the northeastern USA in which pCO_2^{atm} is forced using scenarios from the IPCC Sixth Assessment Report suggest that lake pH could decrease between 0.08 and 0.84 pH units and 0.13–1.31 pH units in the studied lakes depending on whether pCO_2^{atm} attains a level of 600 ppm or 1,100 ppm, respectively, in 2100. These changes translate to mean increases of 31%-65% for the hydrogen ion activity (i.e., a_{H^+}), which compares well with predictions for the oceans. The impact of acidification on indigenous calcifying organism is not currently known but is likely to be detrimental. The acidification of the lakes is also expected to increase the amount of Al complexed with natural dissolved organic ligands (i.e., fulvic acid), but because these lakes are saturated, or nearly so, with respect to gibbsite, Al oxyhydroxide precipitates represent the greatest Al toxicity threat to brook trout (Salvelinus fontinalis). As the lakes acidify over the course of the 21st century, the effects of Al toxicity to brook trout will increase. Consequently, lake acidification by rising anthropogenic pCO₂^{atm} could potentially reverse important gains in water quality that have resulted since implementation of environmental legislation like the Clean Air Act helped to alleviate strong acid deposition and the acidification of freshwater ecosystems from acid rain.

Data Availability Statement

Data employed in this study were compiled from published papers, reports, and on-line sources maintained by several northeastern states. Specifically, the data are available through Boylen et al. (2014), Hintz et al. (2020), Hollocher and Yuskaitis (1993), Solomon et al. (2013), Weiler (1978), the Maine Department of Environmental Protection (2023), the New Hampshire Department of Environmental Services (2022), the Vermont Department of Environmental Conservation, and the New York Department of Environmental Conservation (2019). A summary of the model results is available via Zenodo (Johannesson, 2023a). Example scripts for simulating acidification of Lake Champlain and Lake Superior with version 14.0.01 and higher of the Geochemist's Workbench® software package are also available at Zenodo (Johannesson, 2023b).

References

Abril, G., Bouillon, S., Darchambeau, F., Teodoru, C. R., Marwick, T. R., Tamooh, F., et al. (2015). Technical note: Large overestimation of pCO₂ calculated from pH and alkalinity in acidic, organic-rich freshwaters. Biogeosciences, 12(1), 67-78. https://doi.org/10.5194/bg-12-67-2015

- Alin, S. R., & Johnson, T. C. (2007). Carbon cycling in large lakes of the world: A synthesis of production, burial, and lake-atmosphere exchange estimates. Global Biogeochem Cycles, 21(3), GB3002. https://doi.org/10.1029/2006GB002881
- Andersson, A. J., Mackenzie, F. T., & Bates, N. R. (2008). Life on the margin: Implications of ocean acidification on Mg-calcite, high latitude and cold-water marine calcifiers. Marine Ecology Progress Series, 373, 265-273. https://doi.org/10.3354/meps07639
- Arias, P. A., Bellouin, N., Coppola, E., Jones, R. G., Krinner, G., Marotzke, J., et al. (2021). Technical summary. In V. Masson-Delmonte, P. Zhai, A. Pirani, S. L. Connors, C. Péan, S. Berger, N. Caud, Y. Chen, et al. (Eds.), Climate change 2021: The physical science basis. Contribution of working group I to the Sixth assessment report of the intergovernmental panel on climate change (pp. 33-144). Cambridge University Press. Retrieved from https://www.ipcc.ch/report/ar6/wg1/downloads/report/IPCC_AR6_WGI_TS.pdf
- Balch, W. M., Drapeau, D. T., Bowler, B. C., Record, N. R., Bates, N. R., Pinkham, S., et al. (2022). Changing hydrodynamic, biogeochemical, and acidification properties in the Gulf of Maine as measured by the Gulf of Maine North Atlantic Time Series, GNATS, between 1998 and 2018. Journal of Geophysical Research: Biogeosciences, 127(6), e2022JG006790. https://doi.org/10.1029/2022JG006790

Acknowledgments

This study was supported in part by NSF Grants OCE-2037556 and EAR-2037553 to Johannesson and OCE-1756669 to Burdige. We are very grateful to Dr. Jeremy Deeds (Maine Department of Environmental Protection), David Neils and Kristen Hugger (New Hampshire Department of Environmental Services), Drs. Alison Watts and Robert Cravcraft at the University of New Hampshire, and Dr. Mindy Morales-Williams at the University of Vermont for providing invaluable assistance with data access. Discussions with Dr. Galen McKinley were especially helpful in designing this study, for which we are grateful. The comments of two anonymous reviewers and the editorial handling by Dr. Robinson Fulweiler substantially improved the original manuscript. This paper is dedicated to the memory of Oliver J. White, who was a true friend and is truly missed.

- Baldigo, B. P., George, S. D., Lawrence, G. B., & Paul, E. A. (2020). Declining aluminum toxicity and the role of exposure duration on brook trout mortality in acidified streams of the Adirondack Mountains, New York, USA. *Environmental Toxicology and Chemistry*, 39(3), 623–636. https://doi.org/10.1002/etc.4645
- Beaune, D., Sellier, T., Luquet, G., & Grandjean, F. (2018). Freshwater acidification: An example of an endangered crayfish species sensitive to pH. *Hydrobiology*, 813(1), 41–50. https://doi.org/10.1007/s10750-018-3504-4
- Boylen, C., Eichler, L., Swinton, M., Nierzwicki-Bauer, S., Hannoun, I., & Short, J. (2014). The state of the lake: Thirty years of water quality monitoring on Lake George. Darrin Fresh Water Institute. Retrieved from https://lakegeorgetown.org/assessment-rolls/planning-zoning/plansand-reports/1603-2014-the-state-of-the-lake/file.html
- Burns, D. A., McDonnell, T. C., Rice, K. C., Lawrence, G. B., & Sullivan, T. J. (2019). Chronic and episodic acidification of streams along the Appalachian Trail corridor, eastern United States. *Hydrological Processes*, 34(7), 1498–1513. https://doi.org/10.1002/hyp.13668

Caldeira, K., & Wickett, M. E. (2003). Anthropogenic carbon and ocean pH. Nature, 425(6956), 365. https://doi.org/10.1038/425365a

- Cleveland, L., Little, E. E., Hamilton, S. J., Buckler, D. R., & Hunn, J. B. (1986). Interactive toxicity of aluminum and acidity to early life stages of brook trout. *Transactions of the American Fisheries Society*, 115(4), 610–620. https://doi.org/10.1577/1548-8659(1986)115<610:itoaaa>2.0.co;2
- Cleveland, L., Little, E. E., Ingersoll, C. G., Wiedmeyer, R. H., & Hunn, J. B. (1991). Sensitivity of brook trout to low pH, low calcium and elevated aluminum concentrations during laboratory pulse exposures. *Aquatic Toxicology*, *19*, 303–318.
- Cleveland, L., Little, E. E., Wiedmeyer, R. H., & Buckler, D. R. (1989). Chronic no-observed-effect concentrations of aluminum for brook trout exposed in low calcium, dilute acidic water. In T. E. Lewis (Ed.), *Environmental chemistry and toxicology of aluminum* (pp. 229–246). Lewis Publishers.
- Cole, J. J., Caraco, N. F., Kling, G. W., & Kratz, T. K. (1994). Carbon dioxide supersaturation in surface waters of lakes. Science, 265(5178), 1568–1570. https://doi.org/10.1126/science.265.5178.1568
- Doney, S. C., Fabry, V. J., Feely, R. A., & Kleypas, J. A. (2009). Ocean acidification: The other CO₂ problem. *Annual Review of Marine Science*, 1, 169–192. https://doi.org/10.1146/annurev.marine.010908.163834
- Drever, J. I. (1988). The geochemistry of natural waters (2nd ed., p. 437). Prentice Hall.
- Driscoll, C. T. (1991). Northeast overview. In D. F. Charles (Ed.), Acidic deposition and aquatic ecosystems: Regional case studies (pp. 129–132). Springer-Verlag.
- Driscoll, C. T., Driscoll, K. M., Roy, K. M., & Mitchell, M. J. (2003). Chemical response of lakes in the Adirondack region of New York to declines in acidic deposition. *Environmental Science and Technology*, 37(10), 2036–2042. https://doi.org/10.1021/es020924h
- Driscoll, C. T., Jr., Baker, J. P., Bisogni, J. J., & Schofield, C. L. (1980). Effect of aluminum speciation on fish in dilute acidified waters. *Nature*, 284(5752), 161–164. https://doi.org/10.1038/284161a0
- Fakhraei, H., & Driscoll, C. T. (2015). Proton and aluminum binding properties of organic acids in surface waters of the northeastern U.S. Environmental Science and Technology, 49(5), 2939–2947. https://doi.org/10.1021/es504024u
- Fuss, C. B., Driscoll, C. T., & Campbell, J. L. (2015). Recovery from chronic and snowmelt acidification: Long-term trends in stream and soil water chemistry at the Hubbard Brook Experimental Forest, New Hampshire, USA. *Journal of Geophysical Research: Biogeosciences*, 120(11), 2360–2374. https://doi.org/10.1002/2015jg003063

Gbondo-Tugbawa, S. S., & Driscoll, C. T. (2003). Factors controlling long-term changes in soil pools of exchangeable basic cations and stream acid neutralizing capacity in a northern hardwood forest ecosystem. *Biogeochemistry*, 63(2), 161–185. https://doi.org/10.1023/a:1023308525316

- Gledhill, D. K., White, M. M., Salisbury, J., Thomas, H., Mlsna, I., Liebman, M., et al. (2015). Ocean and coastal acidification off New England and Nova scotia. *Oceanography*, 28, 182–197.
- Gensemer, R. W., Gondek, J. C., Rodriguez, P. H., Arbildua, J. J., Stubblefield, W. A., Cardwell, A. S., et al. (2018). Evaluating the effects of pH, hardness, and dissolved organic carbon on the toxicity of aluminum to freshwater aquatic organisms under circumneutral conditions. *Environmental Toxicology and Chemistry*, 37(1), 49–60. https://doi.org/10.1002/etc.3920
- Gensemer, R. W., & Playle, R. C. (1999). The bioavailability and toxicity of aluminum in aquatic environments. Critical Reviews in Environmental Science and Technology, 29(4), 315–450. https://doi.org/10.1080/10643389991259245
- Golub, M., Desai, A., McKinley, G. A., Remucal, C. K., & Stanley, E. H. (2017). Large uncertainty in estimating pCO₂ from carbonate equilibria in lakes. Journal of Geophysical Research: Biogeosciences, 122(11), 2909–2924. https://doi.org/10.1002/2017jg003794
- Hall, R. J., Driscoll, C. T., Likens, G. E., & Pratt, J. M. (1985). Physical, chemical, and biological consequences of episodic aluminum additions to a stream. *Limnology & Oceanography*, 30(1), 212–220. https://doi.org/10.4319/lo.1985.30.1.0212
- Hanson, P. C., Pollard, A. I., Bade, D. L., Predick, K., Carpenter, S. R., & Foley, J. A. (2004). A model of carbon evasion and sedimentation in temperate lakes. *Global Change Biology*, 10(8), 1285–1298. https://doi.org/10.1111/j.1529-8817.2003.00805.x
- Hintz, W. D., Schuler, M. S., Borrelli, J. J., Eichler, L. W., Stoler, A. B., Moriarty, V. M., et al. (2020). Concurrent improvement and deterioration of epilimnetic water quality in an oligotrophic lake over 37 years. *Limnology & Oceanography*, 65(5), 927–938. https://doi.org/10.1002/ lno.11359
- Hollocher, K., & Yuskaitis, A. (1993). Chemical composition of surface and high-uranium well water, Lake Sunapee area, New Hampshire. Northeastern Geology, 15, 159–169.
- Hunt, C. W., Salisbury, J. E., & Vandemark, D. (2011). Contribution of non-carbonate anions to total alkalinity and overestimation of pCO₂ in New England and New Brunswick rivers. *Biogeosciences*, 8(10), 3069–3076. https://doi.org/10.5194/bg-8-3069-2011
- Ingersoll, C. G., Mount, D. R., Gulley, D. D., La Point, T. W., & Bergman, H. L. (1990). Effects of pH, aluminum, and calcium on survival and growth of eggs and fry of brook trout (Salvelinus fontinalis). Canadian Journal of Fisheries and Aquatic Sciences, 47(8), 1580–1592. https:// doi.org/10.1139/f90-180
- Jagoe, C. H., & Haines, T. A. (1983). Alterations in gill epithelial morphology of yearling Sunapee trout exposed to acute acid stress. *Transactions of the American Fisheries Society*, 112(5), 689–695. https://doi.org/10.1577/1548-8659(1983)112<689:ajgemo>2.0.co;2
- Jane, S. F., Hansen, G. J. A., Kraemer, B. M., Leavitt, P. R., Mincer, J. L., North, R. L., et al. (2021). Widespread deoxygenation of temperate lakes. *Nature*, 594(7861), 66–70. https://doi.org/10.1038/s41586-021-03550-y
- Johannesson, K. H. (2023a). Summary of model results [Dataset]. Zenodo. https://doi.org/10.5281/zenodo.8415182
- Johannesson, K. H. (2023b). Acidification of northeastern USA lakes from rising anthropogenic-sourced atmospheric carbon dioxide and its effects on aluminum speciation [Software]. Zenodo. https://doi.org/10.5281/zenodo.8412170
- Kawahata, H., Fujita, K., Iguchi, A., Inoue, M., Iwasaki, S., Kuroyanagi, A., et al. (2019). Perspective on the response of marine calcifiers to global warming and ocean acidification—Behavior of corals and foraminifera in a high CO₂ world "hot house". *Progress in Earth and Planetary Science*, 6(1), 5. https://doi.org/10.1186/s40645-018-0239-9
- Kleypas, J. A., Buddemeier, R. W., Archer, D., Gattuso, J.-P., Langdon, C., & Opdyke, B. N. (1999). Geochemical consequences of increased atmospheric carbon dioxide on coral reefs. *Science*, 284(5411), 118–120. https://doi.org/10.1126/science.284.5411.118

Lacroix, G. L. (1985). Survival of eggs and Alevins of Atlantic salmon (Salmo salar) in relation to the chemistry of interstitial water in redds in some acidic streams of Atlantic Canada. Canadian Journal of Fisheries and Aquatic Sciences, 42(2), 292–299. https://doi.org/10.1139/f85-037 Lazzarino, J. J., Bachmann, R. W., Hoyer, M. V., & Canfield, D. E., Jr. (2009). Carbon dioxide supersaturation in Florida lakes. Hydrobiologia,

627(1), 169–170. https://doi.org/10.1007/s10750-009-9723-y Likens, G. E., Bormann, F. H., & Johnson, N. M. (1972). *Acid rain* (Vol. 14, pp. 33–40). Environment Science Policy Sustainable Development. Maine Department of Environmental Protection. (2023). Water quality: Lakes. Retrieved from https://www.maine.gov/dep/water/lakes/index.html McKinley, G. A., Fay, A. R., Eddebbar, Y. A., Gloege, L., & Lovenduski, N. S. (2020). External forcing explains recent decade variability of the

- ocean carbon sink. AGU Advances, 1(2), e2019AV000149. https://doi.org/10.1029/2019AV000149 Michelena, T. M., Farrell, J. L., Winkler, D. A., Goodrich, C. A., Boylen, C. W., Sutherland, J. W., & Nierzwicki-Bauer, S. A. (2016). Aluminum toxicity risk reduction as a result of reduced acid deposition in Adirondack lakes and ponds. *Environmental Monitoring and Assessment*, 188(11), 636. https://doi.org/10.1007/s10661-016-5589-4
- Millero, F. J., Woosley, R., Ditrolio, B., & Waters, J. (2009). Effect of ocean acidification on the speciation of metals in seawater. *Oceanography*, 22(4), 72–85. https://doi.org/10.5670/oceanog.2009.98
- Mount, D. R., Swanson, M. J., Breck, J. E., Farag, A. M., & Bergman, H. L. (1990). Responses of brook trout (Salvelinus fontinalis) fry to fluctuating acid, aluminum, and low calcium exposure. Canadian Journal of Fisheries and Aquatic Sciences, 47(8), 1623–1630. https://doi. org/10.1139/f90-184
- National Oceanic and Atmospheric Administration. (2022). Atmospheric CO₂ at Mauna Loa observatory. Global Monitoring Laboratory, Scripps Institution of Oceanography and NOAA. Retrieved from https://gml.noaa.gov/ccgg/trends/
- New Hampshire Department of Environmental Services. (2022). Lake information mapper. Retrieved from https://nhdes.maps.arcgis.com/apps/ webappviewer/index.html?id=1f45dc20877b4b959239b8a4a60ef540
- Ninokawa, A. T., & Ries, J. (2022). Response of freshwater calcifiers to carbon-dioxide-induced acidification. Journal of Marine Science and Engineering, 10(8), 1068. https://doi.org/10.3390/jmse10081068
- O'Reilly, C. M., Sharma, S., Gray, D. K., Hampton, S. E., Read, J. S., Rowley, R. J., et al. (2015). Rapid and highly variable warming of lake surface waters around the globe. *Geophysical Research Letters*, 42(24), 10773–10781. https://doi.org/10.1002/2015gl066235
- Orr, J. C. (2011). Recent and future changes in ocean carbonate chemistry. In J.-P. Gattuso & L. Hansson (Eds.), Ocean acidification (pp. 41–66). Oxford University Press.
- Orr, J. C., Fabry, V. J., Aumont, O., Bopp, L., Doney, S. C., Feely, R. A., et al. (2005). Anthropogenic ocean acidification over the twenty-first century and its impact on calcifying organisms. *Nature*, 437(7059), 681–686. https://doi.org/10.1038/nature04095
- Ou, M., Hamilton, T. J., Eom, J., Lyall, E. M., Gallup, J., Jiang, A., et al. (2015). Response of pink salmon to CO₂-induced aquatic acidification. *Nature Climate Change*, 5(10), 950–955. https://doi.org/10.1038/nclimate2694
- Parkhurst, B. R., Bergman, H. L., Fernandez, J., Gulley, D. D., Hockett, J. R., & Sanchez, D. A. (1990). Inorganic monomeric aluminum and pH as predictors of acidic water toxicity to brook trout (*Salvelinus fontinalis*). *Canadian Journal of Fisheries and Aquatic Sciences*, 47(8), 1631–1640. https://doi.org/10.1139/f90-185
- Peterson, R. H., Daye, P. G., & Metcalfe, J. L. (1980). Inhibition of Atlantic salmon (Salmo salar) hatching at low pH. Canadian Journal of Fisheries and Aquatic Sciences, 37(5), 770–774. https://doi.org/10.1139/f80-103

Phillips, J. C., McKinley, G. A., Bennington, V., Bootsma, H. A., Pilcher, D. J., Sterner, R. W., & Urban, N. R. (2015). The potential for CO₂-induced acidification in freshwater: A great lakes case study. *Oceanography*, 28(2), 136–145. https://doi.org/10.5670/oceanog.2015.37

- Playle, R. C., & Wood, C. M. (1990). Is precipitation of aluminum fast enough to explain aluminum deposition on fish gills? Canadian Journal of Fisheries and Aquatic Sciences, 47(8), 1558–1561. https://doi.org/10.1139/f90-176
- Poléo, A. B. S., Østbye, K., Øxnevad, S. A., Andersen, R. A., Heibo, E., & Vøllestad, L. A. (1997). Toxicity of acid aluminium-rich water to several freshwater fish species: A comparative laboratory study. *Environmental Pollution*, 96(2), 129–139. https://doi.org/10.1016/ s0269-7491(97)00033-x
- Raymond, P. A., Hartmann, J., Lauerwald, R., Sobek, S., McDonald, C., Hoover, M., et al. (2013). Global carbon dioxide emissions from inland waters. *Nature*, 503(7476), 355–359. https://doi.org/10.1038/nature12760
- Santore, R. C., Ryan, A. C., Kroglund, F., Rodriguez, P. H., Stubblefield, W. A., Cardwell, A. S., et al. (2018). Development and application of a biotic ligand model for predicting the chronic toxicity of dissolved and precipitated aluminum to aquatic organisms. *Environmental Toxicology* and Chemistry, 37(1), 70–79. https://doi.org/10.1002/etc.4020

Schindler, D. W. (1988). Effects of acid rain on freshwater ecosystems. *Science*, 239(4836), 149–157. https://doi.org/10.1126/science.239.4836.149 Skjelkvåle, B. L., Stoddard, J. L., Jeffries, D. S., Tørseth, K., Høgåsen, T., Bowman, J., et al. (2005). Regional scale evidence for improvements in

- surface water chemistry 1990–2001. Environmental Pollution, 137(1), 165–176. https://doi.org/10.1016/j.envpol.2004.12.023 Solomon, C. T., Bruesewitz, D. A., Richardson, D. C., Rose, K. C., Van de Bogert, M. C., Hanson, P. C., et al. (2013). Ecosystem respiration: Drivers of daily variability and background respiration in lakes around the globe. Limnology & Oceanography, 58(3), 849–866. https://doi. org/10.4319/lo.2013.58.3.0849
- Stoddard, J. L., Jeffries, D. S., Lükewille, A., Clair, T. A., Dillon, P. J., Driscoll, C. T., et al. (1999). Regional trends in aquatic recovery from acidification in North America and Europe. *Nature*, 401(6753), 575–578. https://doi.org/10.1038/44114
- Strock, K. E., Nelson, S. J., Kahl, J. S., Saros, J. E., & McDowell, W. H. (2014). Decadal trends reveal recent acceleration in the rate of recovery from acidification in the northeastern U.S. Environmental Science and Technology, 48(9), 4681–4689. https://doi.org/10.1021/es404772n
- Tagliarolo, M. (2018). Acidification of aquatic systems. In B. Fath (Ed.), *Encyclopedia of ecology* (2nd ed., Vol. 1, pp. 1–8). Elsevier. https://doi.org/10.1016/B978-0-12-409548.10958-3
- Teien, H.-C., Kroglund, F., Salbu, B., & Rosseland, B. O. (2006). Gill reactivity of aluminum-species following liming. Science of the Total Environment, 358(1–3), 206–220. https://doi.org/10.1016/j.scitotenv.2005.04.023
- Vermont Department of Environmental Conservation and New York Department of Environmental Conservation. (2019). Lake Champlain longterm water quality and biological monitoring program. Waterbury. Retrieved from https://dec.vermont.gov/sites/dec/files/wsm/lakes/docs/ lp_lclongtermprogdesc.pdf
- Waldbusser, G. G., Hales, B., Langdon, C. J., Haley, B. A., Schrader, P., Brunner, E. L., et al. (2015). Saturation-state sensitivity of marine bivalve larvae to ocean acidification. *Nature Climate Change*, 5(3), 273–280. https://doi.org/10.1038/nclimate2479
- Wang, Z., Meador, J. P., & Leung, K. M. Y. (2016). Metal toxicity to freshwater organisms as a function of pH: A meta-analysis. *Chemosphere*, 144, 1544–1552. https://doi.org/10.1016/j.chemosphere.2015.10.032
- Weiler, R. R. (1978). Chemistry of Lake superior. Journal of Great Lakes Research, 4(3-4), 370-385. https://doi.org/10.1016/ s0380-1330(78)72207-0
- Wilkinson, K. J., Bertsch, P. M., Jagoe, C. H., & Campbell, P. C. G. (1993). Surface complexation of aluminum on isolated fish gill cells. Environmental Science and Technology, 27(6), 1131–1138. https://doi.org/10.1021/es00043a012

- Wood, C. M., McDonald, D. G., Ingersoll, C. G., Mount, D. R., Johannsson, O. E., Landsberger, S., & Bergman, H. L. (1990). Effects of water acidity, calcium, and aluminum on whole body ions of brook trout (*Salvelinus fontinalis*) continuously exposed from fertilization to swim-up: A study by instrumental neutron activation analysis. *Canadian Journal of Fisheries and Aquatic Sciences*, 47(8), 1593–1603. https://doi. org/10.1139/f90-181
- Wright, D. A., & Welbourn, P. (2002). Environmental toxicology. In Cambridge Environmental Chemistry Series (Vol. 11, p. 630). Cambridge University Press.

References From the Supporting Information

- Åberg, J., & Wallin, B. (2014). Evaluating a fast headspace method for measuring DIC and subsequent calculation of pCO₂ in freshwater systems. Inland Waters, 4(2), 157–166. https://doi.org/10.5268/iw-4.2.694
- Anderson, G. M., & Crerar, D. A. (1993). Thermodynamics in geochemistry: The equilibrium model (p. 588). Oxford University Press.
- Backes, C. A., & Tipping, E. (1987). Aluminum complexation by an aquatic humic fraction under acidic conditions. *Water Research*, 21(2), 211–216. https://doi.org/10.1016/0043-1354(87)90051-0
- Barnes, R. B. (1975). The determination of specific forms of aluminum in natural waters. *Chemical Geology*, 15(3), 177–191. https://doi.org/10.1016/0009-2541(75)90018-2
- Berton, R., Driscoll, C. T., & Chandler, D. G. (2016). Changing climate increases discharge and attenuates its seasonal distribution in the northeastern United States. *Journal of Hydrology: Regional Studies*, 5, 164–178. https://doi.org/10.1016/j.ejrh.2015.12.057
- Bethke, C. M. (2008). Geochemical and biogeochemical reaction path modeling (2nd ed., p. 543). Cambridge University Press.
- Bethke, C. M., Farrell, B., & Sharifi, M. (2020). The Geochemist's Workbench®, release 14. In *GWB reaction modeling guide* (p. 209). Aqueous Solutions, LLC.
- Browne, B. A., & Driscoll, C. T. (1993). pH-dependent binding of aluminum by a fulvic acid. *Environmental Science and Technology*, 27(5), 915–922. https://doi.org/10.1021/es00042a014
- Cabaniss, S. E. (1990). pH and ionic strength effects on nickel-fulvic acid dissociation kinetics. *Environmental Science and Technology*, 24(4), 583–588. https://doi.org/10.1021/es00074a020
- Cabaniss, S. E., & Shuman, M. S. (1988). Copper binding to dissolved organic matter: II. Variation in type and source of organic matter. Geochimica et Cosmochimica Acta, 52(1), 195–200. https://doi.org/10.1016/0016-7037(88)90067-1
- Campbell, P. G. C., Bisson, M., Bourgie, R., Tessier, A., & Villeneuve, J.-P. (1983). Speciation of aluminum in acidic freshwaters. Analytical Chemistry, 55(14), 2246–2252. https://doi.org/10.1021/ac00264a012
- Cardwell, A. S., Adams, W. J., Gensemer, R. W., Nordheim, E., Santore, R. C., Ryan, A. C., & Stubblefield, W. A. (2018). Chronic toxicity of aluminum, at a pH of 6, to freshwater organisms: Empirical data for the development of international regulatory standards/criteria. *Environ*mental Toxicology and Chemistry, 37(1), 36–48. https://doi.org/10.1002/etc.3901
- Chin, Y.-P., Aiken, G., & O'Loughlin, E. (1994). Molecular weight, polydispersity, and spectroscopic properties of aquatic humic substances. Environmental Science and Technology, 28(11), 1853–1858. https://doi.org/10.1021/es00060a015
- Cole, J. J., Prairie, Y. T., Caraco, N. F., McDowell, W. H., Tranvik, L. J., Striegl, R. G., et al. (2007). Plumbing the global carbon cycle: Integrating inland waters into the terrestrial carbon budget. *Ecosystems*, 10(1), 171–184. https://doi.org/10.1007/s10021-006-9013-8
- Covington, A. K., Whalley, P. D., & Davison, W. (1983). Procedures for the measurement of pH in low ionic strength solutions including freshwater. Analyst, 108(1293), 1528–1532. https://doi.org/10.1039/an9830801528
- Datta, S., Neal, A. W., Mohajerin, T. J., Ocheltree, T., Rosenheim, B. E., White, C. D., & Johannesson, K. H. (2011). Perennial ponds are not an important source of water or dissolved organic matter to groundwaters with high arsenic concentrations in West Bengal, India. *Geophysical Research Letters*, 38(20), L20404. https://doi.org/10.1029/2011GL049301
- DeForest, D. K., Brix, K. V., Tear, L. M., & Adams, W. J. (2018). Multiple linear regression models for predicting chronic aluminum toxicity to freshwater organisms and developing water quality guidelines. *Environmental Toxicology and Chemistry*, 37(1), 80–90. https://doi. org/10.1002/etc.3922

Delany, J. M., & Lundeen, S. R. (1989). The LNLL thermochemical database. Lawrence Livermore National Laboratory. Report UCRL-21658.

- Dickson, A. G. (1993). The measurement of sea water pH. Marine Chemistry, 44(2–4), 131–142. https://doi.org/10.1016/0304-4203(93)90198-w Dickson, A. G., Sabine, C. L., & Christian, J. R. (2007). Guide to best practices for ocean CO₂ measurements (Vol. 3, p. 191). PICES Special Publication.
- Douglas, N. K., & Byrne, R. H. (2017a). Achieving accurate spectrophotometric pH measurements using unpurified meta-cresol purple. Marine Chemistry, 190, 66–72, https://doi.org/10.1016/j.marchem.2017.02.004
- Douglas, N. K., & Byrne, R. H. (2017b). Spectrophotometric pH measurements from river to sea: Calibration of mCP for $0 \le S \le 40$ and 278.15 $\le T \le 308.15$ K. Marine Chemistry, 197, 64–69. https://doi.org/10.1016/j.marchem.2017.10.001
- Driscoll, C. T. (1984). A procedure for the fractionation of aqueous aluminum in dilute acidic waters. *International Journal of Environmental Analytical Chemistry*, *16*(4), 267–283. https://doi.org/10.1080/03067318408076957
- Driscoll, C. T. (1985). Aluminum in acidic surface waters: Chemistry, transport, and effects. *Environmental Health Perspectives*, 63, 93–104. https://doi.org/10.1289/ehp.856393
- Driscoll, C. T., & Postek, K. M. (1996). The chemistry of aluminum in surface waters. In G. Sposito (Ed.), The environmental chemistry of aluminum (2nd ed., pp. 363–418). CRC Press/Lewis Publishers.
- Easley, R. A., & Byrne, R. H. (2012). Spectrophotometric calibration of pH electrodes in seawater using purified m-cresol purple. *Environmental Science and Technology*, 46(9), 5018–5024. https://doi.org/10.1021/es300491s

Eby, G. N. (2004). Principles of environmental geochemistry (p. 514). Thompson, Brooks/Cole.

Environment and Climate Change Canada (2022). Federal environmental quality guidelines—aluminium. In *Canadian Environmental Protection Act, 1999.* Government of Canada Retrieved from https://www.canada.ca/en/environment-climate-change/services/evaluating-existing-substances/federal-environmental-quality-guidelines-aluminium.html

- French, C. R., Carr, J. J., Dougherty, E. M., Eidson, L. A. K., Reynolds, J. C., & DeGrandpre, M. D. (2002). Spectrophotometric pH measurements of freshwater. Analytica Chimica Acta, 453(1), 13–20. https://doi.org/10.1016/s0003-2670(01)01509-4
- Helgeson, H. C. (1968). Evaluation of irreversible reactions in geochemical processes involving minerals and aqueous solutions—I. Thermodynamic relations. *Geochimica et Cosmochimica Acta*, 32(8), 853–877. https://doi.org/10.1016/0016-7037(68)90100-2
- Helgeson, H. C. (1969). Thermodynamics of hydrothermal systems at elevated temperatures and pressures. American Journal of Science, 260, 57-66.

- Helgeson, H. C., Garrels, R. M., & Mackenzie, F. T. (1969). Evaluation of irreversible reactions in geochemical processes involving minerals and aqueous solutions—II. Applications. *Geochimica et Cosmochimica Acta*, 33(4), 455–481. https://doi.org/10.1016/0016-7037(69)90127-6
 Helgeson, H. C., Brown, T. H., Nigrini, A., & Jones, T. A. (1970). Calculation of mass transfer in geochemical processes involving aqueous solutions. *Geochimica et Cosmochimica Acta*, 34(5), 569–592. https://doi.org/10.1016/0016-7037(70)90017-7
- Johannesson, K. H., Palmore, C. D., Fackrell, J., Prouty, N. G., Swarzenski, P. W., Chevis, D. A., et al. (2017). Rare Earth element behavior during groundwater—Seawater mixing along the Kona Coast of Hawaii. *Geochimica et Cosmochimica Acta*, 198, 229–258. https://doi.org/10.1016/j. gca.2016.11.009
- Johnson, M. S., Billett, M. F., Dinsmore, K. J., Wallin, M., Dyson, K. E., & Jassal, R. S. (2010). Direct and continuous measurement of dissolved carbon dioxide in freshwater aquatic systems—Methods and application. *Ecohydrology*, 4, 68–78. https://doi.org/10.1002/eco.95
- Kinniburgh, D. G., Milne, C. J., Benedetti, M. F., Pinheiro, J. P., Filius, J., Koopal, L. K., & van Riemsdijk, W. H. (1996). Metal ion binding by humic acid: Application of the Nica-Donnan model. *Environmental Science and Technology*, 30(5), 1687–1698. https://doi.org/10.1021/ es950695h
- Langmuir, D. (1997). Aqueous environmental geochemistry (p. 600). Prentice Hall.
- LaZerte, B. D. (1984). Forms of aqueous aluminum in acidified catchments of central Ontario: A methodological analysis. Canadian Journal of Fisheries and Aquatic Sciences, 41(5), 766–776. https://doi.org/10.1139/f84-089
- Lake Champlain Long-term Monitoring. (2022). Vermont Department of Environmental Conservation. Retrieved from https://anrweb.vermont.gov/DEC/_DEC/LongTermMonitoringLakes.aspx
- Lofts, S., & Tipping, E. (2011). Assessing WHAM/model VII against field measurements of free metal ion concentrations: Model performance and the role of uncertainty in parameters and inputs. *Environment and Chemistry*, 8(5), 501–516. https://doi.org/10.1071/en11049
- Martin, T. D., Creed, J. T., & Brockhoff, C. A. (1994). Method 200.2, Revision 2.8: Sample preparation procedure for spectrochemical determination of total recoverable elements. U.S. Environmental Protection Agency. Retrieved from https://www.epa.gov/sites/default/files/2015-08/ documents/method_200-2_rev_2-8_1994.pdf
- Millero, F. J. (1979). The thermodynamics of the carbonate system in seawater. Geochimica et Cosmochimica Acta, 43(10), 1651–1661. https:// doi.org/10.1016/0016-7037(79)90184-4
- Pankow, J. F. (2020). Aquatic chemistry concepts (2nd ed., p. 552). CRC Press.
- Pierrot, D. E., Lewis, E., & Wallace, D. W. R. (2006). MS excel program developed for CO₂ system calculations. ORNL/CDIAC-105a. Carbon Dioxide Information Analysis Center, Octory, U.S. Department of Energy. Retrieved from https://cdiac.ess-dive.lbl.gov/ftp/co2sys/ CO2SYS_calc_XLS_v2.1/
- Plummer, L. N., Parkhurst, D. L., & Thorstenson, D. C. (1983). Development of reaction models for ground-water systems. Geochimica et Cosmochimica Acta, 47(4), 665–686. https://doi.org/10.1016/0016-7037(83)90102-3
- Poléo, A. B. S. (1995). Aluminium polymerization—A mechanism of acute toxicity of aqueous aluminium to fish. Aquatic Toxicology, 31(4), 347–356. https://doi.org/10.1016/0166-445x(94)00083-3
- Pötter, L., Tollrian, R., Wisotzky, F., & Weiss, L. C. (2021). Determining freshwater pCO₂ based on geochemical calculation and modelling using PHREEQC. *MethodsX*, 8, 101430. https://doi.org/10.1016/j.mex.2021.101430

Radtke, D. B., Wilde, D. D., Davis, J. V., & Popowski, T. J. (1998). 6.6 Alkalinity and acid neutralizing capacity. In *Techniques of Water-Resources Investigations, Book 9*. U.S. Geological Survey. Retrieved from https://pubs.usgs.gov/twri/twri9a6/twri9a6/twri9a_Section6.6_4-98.pdf

- Raymond, P. A., Caraco, N. F., & Cole, J. J. (1997). Carbon dioxide concentration and atmospheric flux in the Hudson River. *Estuaries*, 20(2), 381–390. https://doi.org/10.2307/1352351
- Rounds, S. A., & Wilde, F. D. (2012). Chapter A6. Section 6.6. Alkalinity and acid neutralizing capacity. In *Techniques of Water-Resources Inves*tigations 09-A6.6 (p. 53). U.S. Geological Survey. Retrieved from https://pubs.usgs.gov/twri/twri9a6/twri9a_6.6.pdf
- Shuman, M. S. (1992). Dissociation pathways and species distribution of aluminum bound to an aquatic fulvic acid. *Environmental Science and Technology*, 26(3), 593–598. https://doi.org/10.1021/es00027a024
- Sutheimer, S. H., & Cabaniss, S. E. (1997). Aluminum binding to humic substances determined by high performance cation exchange chromatography. Geochimica et Cosmochimica Acta, 61, 1–9. https://doi.org/10.1016/s0016-7037(96)00297-9
- Tang, J., & Johannesson, K. H. (2003). Speciation of rare Earth elements in natural terrestrial waters: Assessing the role of dissolved organic matter from the modeling approach. *Geochimica et Cosmochimica Acta*, 67(13), 2321–2339. https://doi.org/10.1016/s0016-7037(02)01413-8
- Tarpey, P. E. (2013). Linking watershed and sub-basin characteristics to in-lake TP concentrations in the Lake Winnipesaukee watershed, New Hampshire, M.S. thesis (p. 181). Plymouth State University. Retrieved from https://summit.plymouth.edu/bitstream/handle/20.500.12774/353/ psu-etd-026.pdf?sequence=1&isAllowed=y
- Thurman, E. M. (1985). Organic geochemistry of natural waters. Nijhoff/Junk.
- Tipping, E. (2005). Modelling Al competition for heavy metal binding by dissolved organic matter in soil and surface waters of acid and neutral pH. *Geoderma*, 127(3–4), 293–304. https://doi.org/10.1016/j.geoderma.2004.12.003
- Tipping, E., Rey-Castro, C., Bryan, S. E., & Hamilton-Taylor, J. (2002). Al(III) and Fe(III) binding by humic substances in freshwaters, and implications for trace metal speciation. *Geochimica et Cosmochimica Acta*, 66(18), 3211–3224. https://doi.org/10.1016/s0016-7037(02)00930-4
- Tipping, E., Lofts, S., & Sonke, J. E. (2011). Humic ion-binding model VII: A revised parameterisation of cation-binding by humic substances. *Environment and Chemistry*, 8(3), 225–235. https://doi.org/10.1071/en11016
- U.S. Environmental Protection Agency. (1983). Methods for chemical analysis of waters and wastes. EPA-600/4-79-020 Retrieved from https:// 19january2017snapshot.epa.gov/sites/production/files/2015-07/documents/epa-6020a.pdf
- U.S. Environmental Protection Agency. (1994). Test methods for evaluating solid wastes. Office of Research and Development. EPA SW-846 (Rev. 0).
- Vachon, D., Prairie, Y. T., & Cole, J. J. (2010). The relationship between near-surface turbulence and gas transfer velocity in freshwater systems and its relationship for floating chamber measurments of gas exchange. *Limnology & Oceanography*, 55(4), 1723–1732. https://doi.org/10.4319/10.2010.55.4.1723
- Weber, W. J., Jr., & Stumm, W. (1963). Buffer systems of natural fresh waters. Journal of Chemical & Engineering Data, 8(3), 464–468. https:// doi.org/10.1021/je60018a062
- Wetzel, R. G. (2001). Limnology: Lakes and river ecosystems (3rd ed., p. 1006). Academic Press.
- Yao, W., & Byrne, R. H. (2001). Spectrophotometric determination of freshwater pH using bromocresol purple and phenol red. Environmental Science and Technology, 35(6), 1197–1201. https://doi.org/10.1021/es001573e